Author’s response to reviews

Title: Telephone-Based Motivational Interviewing versus Usual Care in Primary Care to Increase Physical Activity: A Randomized Pilot Study

Authors:
Deborah Young (Deborah.R.Young@kp.org)
Miki Nguyen (miki.k.nguyen@kp.org)
Ayae Yamamoto (ayae.yamamoto@kp.org)
Magdalena Pomichowsk (Magdalena.E.Pomichowski@kp.org)
Melissa Cornejo (Melissa.Cornejo@kp.org)
Sylvia Paz (Silvia.R.Paz@kp.org)
Karen Coleman (Karen.J.Coleman@kp.org)
Robert Sallis (Robert.E.Sallis@kp.org)
Stephen Fortmann (Stephen.P.Fortmann@kpchr.org)

Version: 2 Date: 26 Nov 2018

Author’s response to reviews:

Dear Sir/Madam,

We are submitting the above-referenced revised manuscript for possible publication in the Pilot and Feasibility Studies. We appreciate the favorable review of the manuscript and that it is potentially acceptable for publication. The revised manuscript shows highlights of the changes made highlighted in yellow. The research represents the results of a pilot study conducted for the sole purpose of establishing feasibility for a future fully powered randomized trial. The work has not been previously published and there is no overlapping information in the manuscript with previously published work. The work will not be submitted for publication elsewhere while under consideration at the Pilot and Feasibility Studies. The study was supported by the National Institutes of Health, R01 DK10090. NIH, nor any of the authors, have any real or perceived conflicts of interest with this research.
All authors take responsibility for the integrity of the work from inception to conclusion. Each of the authors has participated sufficiently in the work to take public responsibility for appropriate portions of the content; all of them have made substantial contributions to all aspects of the research and writing process and have given final approval of the version to be submitted for review to the Pilot and Feasibility Studies.

We look forward to learning the outcome of this submission.